09/14/2020 – EFOG Meeting
	Present: Nena, Shane, Ashley, John, Clay, Bruce, Liz, Bruno

Nena
· Performance appraisals are open until 9/30/2020
· Nena may ask for feedback from others in the group as we have new people in our group and we are not all together due to COVID restrictions
· 11/13/2020 – 12/11/2020 meetings to discuss performance appraisals
· Ops: 911 call for 3 foot snake at 401 ramp this morning
· 911 call last week due to the vapor cloud from the removal of the LN2 tank
· Still working on offer to Adam Stewart
· Accelerator Systems Workshop started today; registration is free; there are a variety of classes
· We are still in Limited Ops; the proposal to move to Limited Ops + is still under review
· Added Wendy back to our Teams chats so that she can see our activity but she is not to respond
· Mohan will talk to sectors 19-24 about FedEx delivery of dewars due to the closure of the parking lot due to LBB
Clay
· Shutdown went well
· Shane suggested that during future shutdowns, we may want to have a 4th FC on-site the last two days of the shutdown
· Startup went well with only a few minor issues
· In the future group should discuss the admin of CCWPs
· Would like to continue our shutdown spreadsheets in Box but they could use some revising
· 2-ID-A shielding verification scheduled for Tues 9/15
· Liz to email bl that they need to approve comm ESAF for this work
· Today alerted by bl staff that labyrinth #4 in 1-ID-B seemed to be sagging/coming apart
· M Henry did some surveys and no leaks were detected
· Solution is to bolt in place a few pieces of Uni-Strut to close the gap
· This can be a temporary repair until we enter the dark period
· Uni-strut is a RSS component; Clay will put on temporary tags; Clay will contact Steve about making permanent tags and updating the CCCL
Ashley
· Today has gone well; mostly ESAFs
· Informed Bruno/John that the O2 monitor in 16-ID-A keeps alarming
· FedEx dewars due to closure of 436 parking lot will be delivered to the 435E corridor; still trying to figure out location(s) for gas yard contents
Shane
· Type Cs at 21, 31, and 22 are done
· Received smokeless powder shipment for 35; will have to check with Bldg 46 to see if material can stay in their safe for the time being
Bruce
· Yu completed one ESAF email request; Bruce to email Yu that the email should be sent when ESAF is submitted not approved
· Bruce to look through the rest of the vector requests to specify when the email should be sent
· Rad Sample Catalog is moving along well
John
· Routine Rad Survey schedule for 2020-3 will be sent out later this week
· 7-ID wants to do their survey this Thursday; need to check with HP on availability
· Will train Steve as his backup when both are on shift together
Bruno
· Will change out ODH sensor at 16-ID-A
· Moved ozone monitor from 28-ID-A to 2-ID-A
· ODH monitors are done except for two that have special access requirements
Liz
· TYSSR for MR-CAT held on Friday, September 11 – review went well
· Clay to update RSS Interface document
· TYSSR for Bio-CAT scheduled for Friday, September 25
· Will contact Sectors this week that are due for TYSSR in October 2020
· Commissioning ESAF file is located in Box
· Pre-Job briefing was held today for Sector 15 so that they can use their Langmuir-Trough in Limited Operations
· These job specific evaluations and/or pre-job briefings will become the norm if groups wish to expand their scope of work beyond what is allowed in Limited Ops
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